**Author's response to reviews**

**Title:** Implementation of a workplace smoking ban in bars: The limits of local discretion

**Authors:**

Theresa Montini (tmontini@aslme.org)
Lisa A Bero (berol@pharmacy.ucsf.edu)

**Version:** 5  **Date:** 19 August 2008

**Author's response to reviews:**

We have made all of the following requested formatting changes. In addition, all of the references are now formatted in Endnote.

Formatting changes requested
-------------------------------

**Major Changes (the author must make these):**

References - References should be cited in the text using consecutive numbers in square brackets. If references [22-24] appear after reference 25, please re-number your references.

Please do not use superscript numbers.

Appendix - This should be placed before the Acknowledgements section of the manuscript.

Table 3 is not a table. Please either format it into a table (details below), or include the information in the manuscript as normal text (please remember to update the citations in the text).

Tables - Tables must be created using a Tables tool. Tables must be divided into rows and columns and be oriented in a portrait page layout.

Footnotes - We are unable to accommodate footnotes in the full text version of the manuscript. We suggest that the footnotes are incorporated into the text of the manuscript. Alternatively, you may wish to include your footnotes in the appendix section.

**Minor Changes (we can make these on the authors’ behalf):**

Country - Please remember to include the country in the affiliation details.

Please remove all figure and table placement text.

Underlining - Please note that underlining is not supported in the final full text; you may wish to change it to bold.
Acknowledgements If you have any acknowledgments, please place an
Acknowledgements section at the end of the manuscript text, directly preceding
the Reference list. Please acknowledge anyone who contributed towards the
study by making substantial contributions to conception, design, acquisition of
data, or analysis and interpretation of data, or who was involved in drafting the
manuscript or revising it critically for important intellectual content, but who does
not meet the criteria for authorship. Please also include their source(s) of
funding. Please also acknowledge anyone who contributed materials essential
for the study.

Please move the Funding section to the Acknowledgements

Table vertical elements - Please note that we are unable to display vertical rules
within the tables nor are we able to write text vertically within the cells - please
re-layout your table without these vertical elements. We are unable to correctly
display merged cells where the merged cell crosses rows.

Please take this opportunity to check your manuscript for any typographical
errors and to make any final corrections or revisions as this is the final proofing
stage for your manuscript, and you will not be able to make changes after
acceptance.

Thank you.